Author's response to reviews

Title: Belt restraint reduction in nursing homes: design of a quasi-experimental study

Authors:

Math JM Gulpers (m.gulpers@zw.unimaas.nl)
Michel HC Bleijlevens (m.bleijlevens@zw.unimaas.nl)
Erik van Rossum (erik.vanrossum@zw.unimaas.nl)
Elizabeth Capezuti (ec65@nyu.edu)
Jan PH Hamers (jph.hamers@zw.unimaas.nl)

Version: 7 Date: 22 February 2010

Author's response to reviews:

February 22th, 2010

Dear editor,

We are pleased that our manuscript entitled “Belt restraint reduction in nursing homes” meets the requirements for publication in BMC Geriatrics.

After you have completed formatting checks of our article, their remains only one thing to be changed. In the current figure 1, four words are highlighted in yellow. This should not be the case and was only done the show what we had changed. Therefore we resubmit a version of figure 1 without this highlight.

Thank you for the opportunity for publication of our paper in BMC Geriatrics.

Yours sincerely, on behalf of the co-authors

Math Gulpers
Maastricht University
School for Public Health and Primary Care (CAPHRI)
Department of Health Care and Nursing Science
P.O. Box 616